Dear Editors and Reviewers,

Thank you for your kind letter on March 10 and for the reviewers’ comments concerning our manuscript entitled “Combined prevention and treatment measures to control nosocomial infections during the COVID-19 pandemic” (Manuscript No: 91286). Those comments are all valuable and very helpful for improving our paper. We have studied comments carefully and have made corrections which we hope meet with approval. The revised portions have been highlighted with yellow color in the revised manuscript.

Here below are our point-by-point responses to the issues raised in the peer-review report(s).

1. Reviewers’ comments:

Reviewer #1:
Scientific Quality: Grade A (Excellent)
Language Quality: Grade A (Priority publishing)
Conclusion: Accept (High priority)
Specific Comments to Authors:
It is very well organized. It has very up-to-date information and is presented in a pragmatic and concise way. It can be a good base of information to prevent future situations of SarsCov2 infections in the hospital environment.
Very interesting article.
Response: Thank you for your positive comments.